Author's response to reviews

Title: Nosocomial acquisition of methicillin-resistant Staphylococcus aureus (MRSA) and extended-spectrum beta-lactamase (ESBL) Enterobacteriaceae in hospitalised patients: a prospective multicenter study.

Authors:

Giulia De Angelis (giulia.deangelis78@gmail.com)
Giovanni Restuccia (giovanni.restuccia@gmail.com)
Silvia Venturiello (silvia.venturiello@gmail.com)
Roberto Cauda (rcauda@rm.unicatt.it)
Surbhi Malhotra-Kumar (surbhi.malhotra@ua.ac.be)
Herman Goossens (herman.goossens@uza.be)
Jacques Schrenzel (jacques.schrenzel@hcuge.ch)
Evelina Tacconelli (etacconelli@rm.unicatt.it)

Version: 3 Date: 22 August 2011

Author's response to reviews:

Dear Diana,

The manuscript “Nosocomial acquisition of methicillin-resistant Staphylococcus aureus (MRSA) and extended-spectrum beta-lactamase (ESBL) Enterobacteriaceae in hospitalised patients: a prospective multicenter study“ has been revised to address the following points.

1. The paper has been formatted according to the journal requirements. Section for authors contributions and Acknowledgment of Funding were added.

2. Trial registration number has been better specified at page 2.

3. Ethic approval was already obtained in all clinical centers. This has been specified at page 3. A copy of the final approval of the coordinator center has been sent to the Editorial office by email.

4. Proof of external funding by the European commission is attached and specified in the manuscript at page 7.

Looking forward to hearing from you at your earliest convenience.

Best regards.

Evelina Tacconelli

Istituto Clinica Malattie Infettive

Università Cattolica
Rome, Italy
email: etaconelli@rm.unicatt.it